Author's response to reviews

Title: Assessment of measles Immunity among Infants in Maputo City, Mozambique.

Authors:

Jagrati V Jani (j.v.jani@medisin.uio.no)
Carol Holm-Hansen (carol.holm-hansen@fhi.no)
TufAria Mussa (tufariamussa@yahoo.com.br)
Arlinda Zango (arlindazango@yahoo.com.br)
Ivan Manhica (ivanmca2004@yahoo.com.br)
Gunnar Bjune (g.a.bjune@medisin.uio.no)
Ilesh V Jani (ivjani@email.com)

Version: 7 Date: 11 November 2008

Author's response to reviews:

Dear editor,

On behalf of my all co-authors, I would like to submit the manuscript entitled "Assessment of Measles Immunity among Infants in Maputo City, Mozambique", as a major article to The BMC Public Health Journal.

Small typographical change in the 4th paragraph under discussion was made from

“We expected that mothers aged 29 years and younger” To “We expected that mothers aged 28 years and younger”.

This version was proofread and approved by all co-authors and we do not have additional changes.

Yours sincerely,

Jagrati Jani
Oslo 11th of November 2008